Notes of the RICH1 Phone meeting 28/10/2005
People Present:

CERN:    
Clara, Fabio, Nick, Olav, Werner(left early)
IC: 

Dave, Gabriel, Trevor
Milano:           Tito

Oxford: 
Neville, Tony(left early)
RAL:               Mike 
Apologies for absence:
IC: Bill
Bristol: Richard
Russia: Vladimir

Discussed Agenda Topics:
1. Magnetic Measurements and Alignment of Magnetic Shield
Two sets of survey measurements have been taken (J. Gayde), before and after the adjustment of the magnetic shield. There results are in EDMS (652642, 673740). The info on how to reference the survey targets to the structure itself is not yet available. Olivier Jamet will provide the info (action Fabio).
The magnetic field measurements have been moved to late Nov. (info form Ann). Power will be available until ~10th Dec.

Side plates: the crane system above RICH1 plus some handling gear is necessary to remove the side plates. Werner will ask Olivier (who is responsible for the craning and lifting) for information about the side door lifting gear and the trolley to remove the doors (action Werner).

2. Gas Enclosure 
2.1 NAB:
Trevor and Tony visited NAB. Both inspected the G.E. and read the survey report. The survey report was far from comprehensive. They asked to have some more survey measurements taken regarding the flatness of the machined surfaces. The survey data is available only on paper.

It seems that the distortions will not be a problem when the G.E. will be constrained in the pit setup. The problem of fixing of the side doors onto the G.E. has been overcome by enlarging the holes. The major distortion of about 6-7mm is at the quartz windows. The angle of the quartz windows changed during the stress relieve, but can live with it. The delivery time to Oxford has still to be discussed: green light for delivery hopefully this or next week. Tony has all ready for the leak tests except for the sniffers.
2.2 Handling Tools:

Trolley: The trolley is close to completion. The final machining is done and needs to be painted. The trolley can not be delivered together wit the G.E. because the vehicle is too small. The trolley can be used also without the mounts.
2.3 Delivery to CERN:
The delivery of the G.E. to CERN is foreseen for early December. It will be delivered to Prevessin to avoid the paper work with the Swiss customs.  The G.E. will be cleaned before it is sent to CERN.  Cleaning using acetone or alcohol would need to be followed by a neutral cleaner (e.g. TEEPOL). Small traces of either could destroy the aerogel (action Tony). The G.E. painting will be done at CERN. Also the paint has to be checked/tested for aerogel compatibility (action Clara). Instead of painting it, the question of black anodizing (previously rejected) should be reopened and this needs to be investigated (action Tony).

3. VELO Seal
The manufacturing is progressing well. The delivery date is week 45. The installation diagrams will be ready for the meeting with Delio. Plan to use matt black paint which resists up to 253 (C.  Erich will test the compatibility of the paint with fluorocarbon gas. Need to revisit the black paint issue. Paint will outgas therefore it is necessary to check the compatibility with aerogel. 
4. Exit Window
The EW looks very good. The spare window is not really needed, therefore it has been decided not to order it. The EW prefitting will be done at Oxford.  The hole will be cut in the EW as soon as the flexible diaphragm/seal design is fixed. 
Diaphragm:  Various silicone (Xo~30cm) mixes are being considered. The drawings for the moulds are being prepared. The silicone will be stretched over the disk of the beampipe. The mould will probably be manufactured outside RAL (quicker). The silicone (Sylguard 186) and fluorocarbon compatibility has been tested some time ago by Erich and no problem was seen. However, once the silicone mix is chosen the silicon and fluorocarbon compatibility test will be redone.
Meeting with beam-pipe team:  15 or 16 Nov at CERN
5. Beryllium Mirrors
5.1 Project:
The overall situation is a cause of considerable concern. There is no schedule available (received following meeting) and contact with the Russian partners is difficult.  Vladimir had technical problems in phoning from Russia and sent an e-mail after the meeting:

  As if on purpose, we have had a severe problem with network, I was unable to enter the web and even to send you a mail. Hope next time it will be better.

Yesterday we had a meeting with "Komposit", so I have some fresh news. 

1. The drawings of the second mirror have reached "Komposit".

2. The story with the second blank is not over, the problem is that because of the long brake in communications between "Komposit" and  "Ulba", the license, which is necessary to send/receive the blank expired. They are working on recommencing of the license and of the contract on 6+1 blanks. I am pushing strongly both E.Vygovsky and V.Golovanov to solve the problem.

Most probably one of them will fly to Kazakhstan next week.

I hope that "Ulba" will not use the situation to revise the cost of the blanks !

3. Meanwhile we are signing the ISTC papers, the main agreement between CERN-IHEP-ISTC is signed, as well as sub-agreement IHEP-Vavilov. We should sign now IHEP-Komposit agreement and it appeared not to be easy, they have a lot of requirements and complains. 

Yesterday there was some progress, I hope that next Tuesday, when they will come to IHEP it will be over. 

4. We are discussing with "Komposit" the way to improve the technology of the annealing. What has been discussed, is to build a new mould from cast iron (the old one was from stainless steel).

                                               Regards,  Vladimir 

5.2 UIR construction drawing:

A signed (by Nick) paper copy (A0) of the UIR drawing was given to Vladimir on the 19/10/05 by Fabio. 

5.3 Metrology: 

The metrology of the prototype beryllium mirror is completed. Soon there will be a metrology document in EDMS.  
5.4 Honeycomb mirrors:  Need to plan for substitute honeycomb mirrors. Agreed to have a meeting to discuss the honeycomb mirrors backup solution on the 3/11/05. 
6. Flat Mirrors
The order for the flat mirrors with a 90m radius of curvature can go ahead (action Carmelo).  Green light has been given by C. Jones.
The polycarbonate test safety ring has been manufactured at IC and has been given to Fabio (October 26th).  A spare RICH2 mirror will be used for the measuring (interferometry) the possible distortions induced by glueing the polycarbonate safety ring to the back of the mirror (action Carmelo+Fabio).

The polycarbonate mounts have been ordered (20/10/05). The delivery is expected for the end of November.
The angle adjuster manufacture at the Bristol workshop is still on hold; the workshop manager refused to give a start date (action Nick).
7. HPD Assembly
Working on the photon funnel and its many interfaces.  Next week will be at CERN to look at the RICH2 HPD column to check model compatibility. Pitch: still open, can wait up to Feb. 2006. If possible use the same pitch as RICH2.
8. Quartz Windows

Go ahead and prepare paperwork for order. It takes ~7 months to get the quartz windows from the launch of the order. A drawing will be prepared by Trevor.
9. RICH1 Status Review: end of Nov05
10. Next meeting
Friday 11th November at 10:00(GMT) 11:00(CET). 
